

Global Biogeochemical Cycles[®]

RESEARCH ARTICLE

10.1029/2022GB007577

Key Points:

- High resolution data indicated that photosynthesis and respiration mainly drove pH and dissolved O₂ (DO) daily variability in reef systems
- Diel ranges in pH and DO and their co-variation may be linked to reef metabolism and benthic ecosystem composition

Supporting Information:

Supporting Information may be found in the online version of this article.

Correspondence to:

S. E. Cryer, s.e.cryer@soton.ac.uk

Citation:

Cryer, S. E., Evans, C., Fowell, S. E., Andrews, G., Brown, P., Carvalho, F., et al. (2023). Characterizing reef net metabolism via the diel co-variation of pH and dissolved oxygen from high resolution in situ sensors. *Global Biogeochemical Cycles*, *37*, e2022GB007577. https://doi. org/10.1029/2022GB007577

Received 2 SEP 2022 Accepted 23 AUG 2023

Author Contributions:

Conceptualization: Sarah E. Cryer, Claire Evans, Richard Sanders, Arlene Young, Socratis Loucaides Data curation: Sarah E. Cryer, Socratis Loucaides Formal analysis: Sarah E. Cryer, Filipa Carvalho Funding acquisition: Sarah E. Cryer, Claire Evans, Richard Sanders, Arlene Young, Socratis Loucaides Investigation: Sarah E. Cryer, Gilbert Andrews, Diana Degallerie, Samir Rosado, James A. Strong, Derrick Theophille, Socratis Loucaides

© 2023. The Authors.

This is an open access article under the terms of the Creative Commons Attribution License, which permits use, distribution and reproduction in any medium, provided the original work is properly cited.

Characterizing Reef Net Metabolism Via the Diel Co-Variation of pH and Dissolved Oxygen From High Resolution in Situ Sensors

Sarah E. Cryer^{1,2}, Claire Evans², Sara E. Fowell², Gilbert Andrews³, Peter Brown², Filipa Carvalho², Diana Degallerie⁴, Jake Ludgate², Samir Rosado³, Richard Sanders^{2,5}, James A. Strong², Derrick Theophille⁴, Arlene Young³, and Socratis Loucaides²

¹Ocean and Earth Science, University of Southampton, Southampton, UK, ²National Oceanography Centre, Southampton, UK, ³Coastal Zone Marine Authority and Institute, Belize City, Belize, ⁴Dominica Fisheries Division, Roseau, Commonwealth of Dominica, ⁵NORCE Norwegian Research Centre, Bergen, Norway

Abstract Coral reefs are subject to degradation by multiple environmental stressors which are predicted to intensify. Stress can alter ecosystem composition, with shifts from hard coral to macroalgae dominated reefs often accompanied by an increase in soft corals and sponges. Such changes may alter net ecosystem metabolism and biogeochemistry by shifting the balance between photosynthesis, respiration, calcification and dissolution. We deployed high temporal resolution pH and dissolved oxygen (DO) sensors at four Caribbean reef sites with varying covers of hard and soft corals, sponges and macroalgae. The resultant data indicated that the strength of the "metabolic pulse", specifically the co-variation in daily pH and DO oscillations, was driven by the net balance of light -dependent and -independent metabolism. pH and DO were positively correlated over the diel cycle at coral dominated sites, suggesting that photosynthesis and respiration were the major controlling processes, and further indicated by agreement with a simple production:respiration model. Whereas, at a site with high macroalgal cover, pH and DO decoupling was observed during daylight hours. This indicates that an unidentified light-driven process altered the expected pH:DO relationship. We hypothesize that this could be mediated by the higher levels of macroalgae, which either stimulated bacterial-mediated carbonate dissolution via the production and release of allelopathic compounds or retained oxygen, evolved during photosynthesis, in the gaseous form in seawater (ebullition). Our work demonstrates that high resolution monitoring of pH and DO provides insight into coral reef biogeochemical functioning and can be key for understanding long-term changes in coral reef metabolism.

Plain Language Summary Coral reefs are regarded as the rainforests of the sea, supporting 25% of marine species and millions of people worldwide. Yet, they are facing a number of threats including rising sea surface temperature, decreasing ocean pH, overfishing and increased competition with algae. As coral reefs experience stress, the relative abundance of hard coral, algae, soft corals and sponges can change. A healthy reef is generally thought of as one with a high proportion of hard coral as their skeletons form the backbone of the reef environment. Under stress, however, a reef can transition from hard coral to algae dominated with significant implications for the entire reef ecosystem. In this paper, we suggest that the reef metabolism, which is driven by benthic composition, drives the natural variability and the relationship between pH and dissolved oxygen (DO) over a 24-hr cycle. Our data suggest that the daily range in pH and DO is larger at sites with more hard coral and less algae, and the relationship between pH and DO is stronger. High resolution measurements of pH and DO using autonomous sensors could potentially reveal vital information on ecosystem functioning and may be valuable tools in monitoring changes in coral reefs.

1. Introduction

Coral reefs are subject to numerous intensifying environmental stressors, including rising sea surface temperatures and ocean acidification, and anthropogenic activities, such as overfishing and nutrient loading (Bove et al., 2020; Bozec et al., 2016; Pendleton et al., 2016; Romanó de Orte et al., 2021; Silbiger et al., 2014, 2018). This has resulted in a global decline of coral reef health, with an approximate 50% reduction in coral cover from 1957 to 2007 (Eddy et al., 2021). A key indicator of declining reef health is changes to ecosystem community composition, which frequently results in phase shifts. Phase shifts are often characterized by a change in



Methodology: Sarah E. Cryer, Sara

E. Fowell, Gilbert Andrews, Diana

Degallerie, Jake Ludgate, Samir Rosado,

James A. Strong, Derrick Theophille, Arlene Young, Socratis Loucaides

Project Administration: Claire Evans.

Sanders, James A. Strong, Derrick

Theophille, Arlene Young, Socratis

Derrick Theophille, Arlene Young,

Validation: Jake Ludgate, Socratis

Visualization: Sarah E. Cryer, Filipa

Writing - original draft: Sarah E. Cryer,

Writing - review & editing: Sarah E.

Filipa Carvalho, Richard Sanders, James

A. Strong, Derrick Theophille, Socratis

Cryer, Sara E. Fowell, Peter Brown,

Loucaides

Socratis Loucaides

Socratis Loucaides

Socratis Loucaides

Loucaides

Carvalho

Claire Evans

Loucaides

Sara E. Fowell, Diana Degallerie, Richard

Resources: Claire Evans, Sara E. Fowell,

Gilbert Andrews, Diana Degallerie, Jake Ludgate, Samir Rosado, James A, Strong,

Software: Sara E. Fowell, Jake Ludgate.

Supervision: Claire Evans, Peter Brown, Filipa Carvalho, Richard Sanders, dominance from hard coral to macroalgae (Hughes, 1994; Nyström et al., 2008). Healthier reefs are generally associated with higher hard coral cover and lower macroalgae presence (Mumby et al., 2007). Following coral mortality, algal assemblages are usually the first colonizers of the bare substrate (Romanó de Orte et al., 2021), and once established, macroalgae can damage adult corals and suppress coral recruitment (Brooker et al., 2016). Phase shifts to the dominance of other life forms, including soft corals and sponges, can also occur following a disturbance to a reef (Norström et al., 2009). Sponge abundance on reefs, for example, has risen globally (Bennett et al., 2017), with an increase in bioeroding clionid sponges reported in the Caribbean (Antonius & Ballesteros, 1998; López-Victoria & Zea, 2005). Coral loss is particularly acute in the Caribbean where hard coral cover has declined by 80% on average since the 1970s (Gardner, 2003). Hard corals form aragonite skeletons, which typically provide the physical framework for a reef; therefore, their loss can have profound consequences, not only for the coral (Nyström et al., 2008) but also for the entire reef ecosystem and local biogeochemistry (Webb et al., 2021).

Calcification is the key process responsible for constructing the physical reef (Erez et al., 2011), and to maintain their habitat, reefs need to calcify at a rate greater than the combined rates of erosion, dissolution (Andersson & Gledhill, 2013; Erez et al., 2011; Kleypas et al., 1999) and sea-level rise (Perry et al., 2018). It may, therefore, be more appropriate to define a healthy reef as one that is actively growing and accreting net calcium carbonate (CaCO₃), as opposed to the proportions of different benthos (Smith et al., 2016). Reductions in rates of calcification as a result of anthropogenic stressors such as ocean acidification (Erez et al., 2011), rising sea surface temperatures (D'Olivo & McCulloch, 2017), or eutrophication (Steiner et al., 2018) could result in reef-wide consequences due to alteration of reef structure (Hoegh-Guildberg et al., 2007; Kleypas et al., 1999). Ocean acidification, caused by the air-to-sea flux of atmospheric carbon dioxide (CO₂), is of particular concern for marine calcifiers due to the consequent decrease in carbonate ion concentrations [CO₃^{2–}] (Shamberger et al., 2018) and therefore calcite and aragonite saturation states. As calcification at the organism level responds mainly to [CO₃^{2–}] (Erez et al., 2011), community calcification is expected to decline as the ocean becomes more acidic (e.g., Bove et al., 2019; Shamberger et al., 2018).

The extent to which biology affects seawater chemistry in shallow reef waters is determined by the coral reef community metabolism (Langdon et al., 2003; Price et al., 2012; Smith et al., 2013), topography (depth, open water connectivity, etc.) and a number of physical processes (i.e., mixing, currents, etc) (Falter et al., 2013). During daylight, photosynthesis by autotrophs produces oxygen (O_2) , while consuming CO_2 , increasing seawater pH and shifting carbonate speciation toward higher $[CO_3^{2-}]$ (Suzuki et al., 1995). In the absence of light, respiration dominates and, therefore, dissolved oxygen (DO) is consumed and CO₂ released, reducing pH and $[CO_3^{2-}]$. In non-calcifying photoautotrophs such as algae, the relationship between DO production/consumption and pH is reasonably predictable from the stoichiometric "Redfield Ratio", assuming the change in total alkalinity (TA) is negligible (Smith et al., 2013). Corals, however, have a more complex pH to DO relationship given that they calcify, photosynthesize (by zooxanthellae symbionts) and respire, thereby altering seawater dissolved inorganic carbon (DIC) and TA. During calcification TA and DIC are removed from seawater at a 2:1 ratio and hydrogen ions (H⁺) are released causing an overall decline in pH (Jokiel et al., 2014; Wolf-Gladrow et al., 2007). CaCO₃ dissolution releases TA and DIC at a 2:1 ratio, therefore increasing pH (Wolf-Gladrow et al., 2007). Deviations from a "Redfield" pH:DO covariation could, therefore, be evidence of CaCO₃ calcification or dissolution, as neither process is directly related to O₂ metabolism. Calcification occurs during both day and night, however its rates are higher during daylight when photosynthesizing coral symbionts can enhance calcification rates through light-enhanced calcification (Gattuso et al., 1999; Moya et al., 2006). This algal-coral symbiosis is also referred to as the "biological engine producing the reef" (Roth, 2014). Dissolution typically happens at night (e.g., Moav-Barzel et al., 2023; Page et al., 2016; Romanó de Orte et al., 2021) as a consequence of decreasing pH and $[CO_3^{2-}]$, which results from CO₂ generated by respiration no longer being consumed by photosynthesis. Net reef dissolution can be an indicator of reef ecosystem health decline, and has been associated with vulnerability to environmental stress (Hoegh-Guildberg et al., 2007). Carbonate chemistry speciation in reef ecosystems is, therefore, dependent on the community-specific balance between calcification, dissolution, respiration and photosynthesis, which is a product of the relative abundance of calcifying, non-calcifying (Anthony et al., 2011), autotrophic and heterotrophic organisms.

Biogeochemical monitoring can shed light on marine ecosystem functioning by revealing natural environmental variability and providing baselines against which long-term changes can be assessed. Typically, ecosystems exhibit threshold responses to stress, rather than responding linearly to environmental change (Hughes et al., 2010); this

Global Biogeochemical Cycles

10.1029/2022GB007577



Figure 1. (a) Technical drawing of the fixed-point observing platform with lab-on-chip pH sensor, rechargeable Ni-MH battery pack and Seabird CT-DO sensor labeled. (b) Platform deployed on the Goff's Caye fore reef site.

could lead to changes in seawater biogeochemistry being detecatable before changes are identifiable by visual ecosystem assessments. High temporal resolution biogeochemical observations from coral reefs are lacking. The shallow bathymetry and inaccessibility of coral reefs make it difficult to collect high resolution data through the traditional means of discrete water sampling. These barriers can be overcome by autonomous sensors and technological advances (e.g., Cryer et al., 2020; Schaap et al., 2021; Takeshita et al., 2018; Yin et al., 2021), which enable sustained high-temporal resolution and robust measurement. Over the last decade or so, as sensor technology has advanced, more high-resolution reef pH data sets have become available (e.g., Cyronak et al., 2020; Kline et al., 2015; McMahon et al., 2018; Santos et al., 2011; Silbiger et al., 2014). The availability of carbonate sensors (particularly TA and DIC) capable of high-quality measurements in dynamic environments, such as coral reefs, remains limited. Therefore, data from reefs, especially coupled pH and DO measurements (e.g., Gray et al., 2012; Pezner et al., 2021; Takeshita et al., 2021; Takeshita et al., 2012; Pezner et al., 2021; Takeshita et al., 2018), remain scarce.

Stress induced disturbances to the balance of metabolic processes on reefs and the resultant phase shifts, are likely to be evident in the diel cycles of biogeochemical parameters such as pH and DO concentrations. In this work we investigated the relationship between biogeochemical parameters and benthic composition on four Caribbean reef sites, via high-resolution, simultaneous measurements of pH and DO and their co-variation. We propose that the magnitude of the diel cycle of pH and DO and the relationship between them can be indicative of the reef's "metabolic pulse" analogous to that described by Cyronak et al. (2018) for DIC and TA.

2. Materials and Methods

2.1. Data Collection

Custom-built fixed-point observing platforms (Figure 1) were deployed at four sites on Caribbean coral reefs (see Section 2.3). The fixed-point observatories were developed at the National Oceanography Centre, UK (Ocean Technology and Engineering group) to enable high performance, user friendly and low-cost ocean acidification observations in coastal waters as part of the Commonwealth Marine Economies Program (https://cmeprogramme.org). Their relatively small size and low weight in air (<50 kg) enable deployment and recovery by hand in coastal and shallow waters without the need of a winch or a large vessel. Each platform includes a microfluidic lab-on-chip (LOC) pH sensor (Yin et al., 2021) and a pumped conductivity, temperature and dissolved oxygen (CT-DO) sensor (SBE 37 SMP ODO MicroCAT; Seabird Scientific, Bellevue WA, USA). The technology used was designed to assist developing nations to monitor their coastal waters in response to the UN Sustainable Development Goals. Communication between the two sensors and power supply are enabled through a custom-built communications hub. The hub enabled the pH sensor to operate the CT-DO sensor on demand and use its temperature and reference salinity (psu) measurements to calculate total scale pH (pH_T) as per Liu et al. (2011), at in situ conditions. From here onward quoted pH values are on the total scale. Power was provided by a custom-made rechargeable Ni-MH battery pack enclosed within a polyvinyl chloride (PVC) 100 m depth rated housing. The battery provides enough power for hourly meas-

Characteristics of Deployment Locations

Site	Country	Latitude	Longitude	Site description	Depth (m)
Goff's Caye Fore Reef	Belize	17.35229°N	88.03381°W	Dominated by hard and soft corals	6
Goff's Caye Back Reef	Belize	17.35401°N	88.03789°W	Dominated by rubble, sand and soft coral	4.5
English Caye	Belize	17.326094°N	88.04933°W	Dominated by macroalgae, soft coals and sponges. Back reef environment	6.5
Soufrière	Dominica	15.236866°N	61.370282°W	Mostly dead coral and rock with encrusting species (corals, sponges and algae) beginning to grow. Back reef environment	9

urements over a 1-month deployment. The sensors and battery pack were contained within a stainless-steel frame designed for seabed or moored deployments. Data generated were stored on-board the pH sensor, while optional near-real time data telemetry was enabled by Trident Sensors Iridium data and tracking beacon fixed to a surface buoy and connected to the data hub by a 25 m cable. The LOC pH sensor (now manufactured and sold by ClearWater Sensors Ltd.) measures pH on the total scale using purified meta-Cresol Purple indicator dye as described in Clayton and Byrne (1993). It has a combined standard measurement uncertainty of <0.010 at pH 8.0 (<0.009 at pH 8.5 and <0.014 at pH 7.5) and measurement reproducibility of <0.001 pH units (Yin et al., 2021). Measurement frequency can be adjusted to a maximum of 1 measurement every 8 min.

2.2. Collection and Analysis of DIC and TA

Two DIC and TA samples were collected at the Belize Goff's Caye sites (Section 2.3) using a 5 L Niskin bottle deployed to sensor depth on 4 November 2019. Samples were then collected in 250 ml borosilicate glass bottles, poisoned with 50 μ L saturated mercuric chloride solution, sealed with Apiezon L silicone grease on ground-glass stoppers and stored in the dark at room temperature before transport to, and analysis in, the UK. TA was determined using a two-stage potentiometric open-cell titration using a Scripps TA Titration System (Dickson et al., 2003). The titration acid was calibrated each day using CRMs (Certified Reference Material, Dickson Lab Batches 180 and 191; Dickson, 2010), with an average uncertainty on CRMs of $\pm 1.7 \mu$ mol/kg. DIC was measured by coulometry using a MARIANDA VINDTA 3C system (Mintrop, 2004). CRMs (Batches 180 and 191; Dickson, 2010) were run every ~12 samples, with an average uncertainty of $\pm 3.0 \mu$ mol/kg.

2.3. Study Sites

Observing platforms were deployed on the Meso-American Barrier Reef (Belize) and the fringing reef of Dominica (Table 1). We collected high resolution time series data from four different sites: three sites in Belize (fore reef and back reef at Goff's Caye, back reef at English Caye, Figure 2) and a single back reef site in Dominica (5 separate deployments near Coral Gardens; Figure 2).

2.3.1. Belize

Two observing platforms were deployed simultaneously for 10 days at two sites on the Belize Barrier Reef by Goff's Caye (Figure 2a) in October–November 2019, around 20 km offshore of Belize City. Wet season is typically June–November; however, during our deployment Belize was experiencing a prolonged dry season. Coral cover in Belize is estimated to have declined from 26.3% to 10.7% from 1997 to 2016, while macroalgal cover increased from 12.9% to 39.7% (Alves et al., 2022). Despite Belize putting measures in place to tackle reef degradation, such as the implementation of Marine Protected Areas and a ban on herbivorous fishing, coral cover is yet to show signs of recovery (Alves et al., 2022; Cox et al., 2017). However, recent studies suggest coral cover has not decreased further in the last decade (Mumby et al., 2021). Extensive seagrass beds characterized the seabed between the mainland and the Belize Barrier Reef.

One platform was deployed on the fore reef (6 m depth) and the other on the back reef (4.5 m depth) in order to compare short-term pH and DO variability between the two sites simultaneously. Prior to the deployment, the two observing platforms were placed on the seabed adjacent to each other for two hours at a 20-min measurement frequency in order to determine any measurement offsets between the sensors. The pH difference was calculated at 0.016, which is within the cumulative uncertainty of the two sensors (each sensor has an estimated measure-





ment uncertainty of ~0.01), DO difference was 2.95 μ mol/L (sensor accuracy $\pm 3 \mu$ mol/L), 0.003°C difference in temperature (accuracy ± 0.002 °C) and a 0.008 g/kg difference in salinity (accuracy $\pm 1 \times 10^{-4}$ g/kg at 25°C). One of the two platforms was redeployed afterward at a back reef site adjacent to English Caye for long-term monitoring. The English Caye site, 3.3 km south of Goff's Caye, was chosen mainly due to the safety provided by the continuous presence of the caye's lighthouse keeper.

2.3.2. Dominica

An observing platform was deployed at a back reef site within the Soufriere-Scott's Head Marine Reserve, around 100 m offshore (Figure 2c). The platform was deployed five times for 11–16 days in 2018 (March, April, June) and in 2019 (March and July), covering both wet and dry seasons. Wet season occurs between June and October, with heaviest rainfall typically between late summer and fall and peak hurricane season late August–early September. Coral reefs in Dominica are comparatively isolated (Steiner, 2015), consisting of a rock base and are predominately sponge-dominated reefs (Madisetti, 2018). In September 2017 Hurricane Maria, a Category 5 hurricane, hit Dominica and destroyed an estimated 20%–40% of the reefs across the island (Madisetti, 2018). Coral Gardens, the closest site surveyed to the platform's deployment, was extensively damaged, with trees, rocks and silt from the associated landslides smothering the reef (Madisetti, 2018). At Champagne Reef, everything shallower than 3 m water depth was completely destroyed, while the reef wall was damaged to a depth of 15 m and many large sponges were destroyed (Madisetti, 2018).

2.4. Characterization of Benthos

Benthic ecosystem assemblages around each observatory were characterized from photographs of the seabed. In Belize, photos were taken using a GoPro Hero 3+ by scuba divers or snorkelers immediately after platform deployment. These photos were within 10 m of the sensor frame. In Dominica, a GoPro mounted on a mini Remotely Operated Vehicle (DeepTrecker) was deployed over the side of the boat and screengrabs were taken from recorded video. The photos from Dominica were taken on 10 March 2019, 1 year after the first deployment but before deployments 4 and 5.

Benthos and substrate were determined using Coral Point Count (CPC) with Excel Extensions (Kohler & Gill, 2006). Photos, representative of the reef environment, were loaded into CPC, 4 for each of the Belize sites and 13 screengrabs for Dominica. For each photo an 8×8 grid was constructed and within each cell a point was randomly distributed. The marine benthos or substratum for each point was then identified manually.

2.5. Statistical Analysis & Production Respiration (P:R) Model

A two-sample *t*-test (Matlab, 2018) was used to determine if pH, DO, salinity and temperature differences between Belize fore reef and back reef sites were statistically significant, with significance determined if $\rho < 0.05$. Pearson's Correlation (Matlab, 2022) was used to test the relationship between pH and DO or between salinity and temperature, at each site. A one-way anlaysis of variance (ANOVA) was used to test the differences in pH or DO between Dominica deployments.

pH and DO observations were compared against theoretical Photosynthesis: Respiration (P:R) model, following Frieder et al. (2012). This model describes the expected relationship between pH and DO in systems when photosynthesis and respiration are assumed to be the dominant controlling processes. For a given change in DO (Δ DO, in mol kg⁻¹), proportional changes in dissolved CO₂ (Δ CO₂, equal to Δ DIC, also in mol kg⁻¹) are expected according to the stoichiometric "Redfield" ratio Δ CO₂ = $\frac{\Delta O_2}{-159/106}$ (Anderson, 1995). The model assumes limited contributions from air-sea gas fluxes, which is a reasonable approximation given that maximum flux estimates around Belize are ~5 mol m⁻² yr⁻¹ (Wanninkhof et al., 2020) and calcification or carbonate dissolution if prevalent would be observed through pH divergence without an accompanying oxygen change.

 Δ DIC was converted to changes in pH (Δ pH) using internal inorganic carbon thermodynamic constants in CO2SYS (Lewis & Wallace, 1998; Sharp et al., 2020). For this we use TA (TA = TA_i + Δ TA; where *i* indicates initial concentrations, ΔTA = change in TA) and DIC (DIC = DIC_i + ΔDIC) as input parameters, and dissociation constants for carbonic acid from Mehrbach et al. (1973) refitted by Dickson and Millero (1987), for KSO₄ from Dickson (1990), for total Boron concentration from Lee et al. (2010), $KH_{\rm F}$ from Perez and Fraga (1987) and median temperature (°C) and salinity (psu) from each site (Table 2). Initial TA (TA.) for each site was calculated from salinity using a TA-salinity parameterization for the North Atlantic and Caribbean region (TA = $57.3 \times \text{salinity} + 296.4$; Cai et al., 2010). Δ TA was calculated using $\Delta TA = \frac{\Delta CO_2}{-^{15/106}}$ (Frieder et al., 2012; Stumm & Morgan, 1996), which accounts for changes in seawater proton concentration [H⁺] during photosynthesis/respiration. Initial DIC (DIC_i) was adjusted to match as much as possible the P:R model to our observations. Adjusted DIC, values were 2,075.9 µmol/kg (Goff's Caye fore reef), 2,052.5 µmol/kg (Goff's Caye back reef), 2,078.6 µmol/kg (English Caye), and 2,041.2 µmol/kg (Dominican site). Measured DIC and TA from samples collected at Goff's Caye fore reef (DIC: 2,020 µmol/kg; TA: 2,352 µmol/ kg) and back reefs (DIC: 2,026 µmol/kg; TA: 2,345 µmol/kg) on 4 November 2019, were within the DIC and TA ranges used for the Goff's Caye fore reef (DIC: 1,822-2,133 µmol/kg, TA: 2,344-2,397 µmol/kg) and back reef (DIC: 1,798–2,109 µmol/kg, TA: 2,344–2,396 µmol/kg) P:R models. We tested the validity of our choice of TA; and DIC; by varying it across typical oceanic ranges (Figure S3 in Supporting Information S1, P:R model calculated using a range of TA; and DIC;). We found that it did not affect our data interpretation due to it primarily affecting only the model's y intercept and not the relationship calculated between pH and DO.

3. Results

3.1. Diel Cycle in pH and DO

The amplitude of the diel cycle in both pH and DO varied between sites (Figures 3–6), with the minimum and maximum measurements recorded summarized in Table 2. Within a 24-hr period, the maximum diel range in pH at each site was 0.116 (Goff's Caye fore reef), 0.137 (Goff's Caye back reef), 0.164 (English Caye), and 0.097 (Dominica—March 2018 deployment). Minimum pH was observed around dawn (04:00–06:00, local time) at all sites with maximum pH between 11:00–14:00. Minimum DO was observed between 05:00 and 07:00 with maximum DO between 13:00–15:00 and maximum DO either coincided with maximum pH (Goff's Caye fore reef and back reef) or was observed shortly after (English Caye and Dominica). The median range in pH and DO was calculated using the median of the daily minimum and median of the daily maximum. The largest median range in both pH and DO was observed at the Goff's Caye back reef site (Table 2, Figures 3c and 3d), with the minimum range recorded at the Dominica site (Table 2, Figures 3g and 3h). In Dominica, median DO and pH varied significantly between the five deployments (DO; one way ANOVA, p = 0.0422 and pH; one way ANOVA, p < 0.0001).

3.2. Intra-Reef Comparison (Goff's Caye, Belize)

Figure 4 shows pH, DO, temperature and absolute salinity over the 9.5-day deployment at the fore and back reef sites (Goff's Caye). The variability in all parameters was similar between the back and fore reef sites, although



Global Biogeochemical Cycles

19449224, 2023, 9, Downloaded from https://agupubs.onlinelibrary.wiley.com/doi/10.1029/2022GB007577 by University Of Southampton, Wiley Online Library on [1809/2023]. See the Terms

and Conditions (https://onlinelibrary.wiley.com/terms

Maximum, Mir	nimum, Median and Range oj	f pH, DC	, Absolute	Salinity a	nd Tempe	rature Mea	surements	s From Each	ı Deploymen	t					
		Ŋ	Sampling interval	Max	Min	Median htt	Median	Mav DO	Min DO	Median DO range	Median	Salinity	Median	Tennerature	Median
Site	Date	days	(hour)	pH _T	pH_{T}	range	pH _T	(J/Iomul)	(J/Iomul)	(µmol/L)	(J/Iomul)	(g/kg)	(g/kg)	range (°C)	(°C)
Fore Reef, Belize	25 October 2019–4 November 2019	9.5	1	7.991	7.849	0.066	7.937	206.7	131.3	30.7	176.1	0.53	36.08	0.69	29.61
Back Reef, Belize	25 October 2019–4 November 2019	9.5	1	8.024	7.8820	0.070	7.968	217.4	144.5	32.0	180.8	0.66	36.06	0.97	29.59
English Caye, Belize	06 November 2019–2023 December 2019	47	7	8.090	7.866	0.047	7.965	196.5	137.5	11.9	174.6	1.29	36.42	4.20	28.71
Soufrière, Dominica	15 March 2018–2029 March 2018	14	7	8.066	7.969	0.022	8.021	209.2	194.2	5.6	202.0	0.99	35.80	1.02	27.13
Soufrière, Dominica	9 April 2018–2020 April 2018	11	7	8.035	7.963	0.026	8.007	210.5	194.2	7.6	202.3	1.45	35.39	0.65	27.52
Soufrière, Dominica	08 June 2018–2024 June 2018	16	2	8.029	7.95	0.032	8.006	208.8	190.9	9.4	200.0	1.72	34.40	0.19	27.87
Soufrière, Dominica	27 March 2019–07 April 2019	11	2	8.059	8.004	0.024	8.034	208.5	194.6	7.5	202.6	0.35	36.40	0.71	26.77
Soufrière, Dominica	03 July 2019–14 July 2019	Ξ	7	8.081	7.973	0.03	8.023	206.7	191.2	8.1	201.1	0.56	34.53	0.46	28.35

Table 2





Figure 3.





Figure 4. Time series for a 9.5 days deployment showing (a) pH_T , (b) DO, (c) temperature and (d) absolute salinity for the Goff's Caye fore reef site (purple, solid line) and back reef (pink, dashed line). Nighttime is shown by gray shading. The corresponding box plots for each site are also plotted in purple for fore reef and pink for back reef.

pH and DO values recorded at the back reef were significantly higher (two sample *t*-test, pH p < 0.0001, DO p < 0.0001) with medians 0.029 and 6.23 µmol/L higher respectively. The larger amplitudes in pH and DO were observed when salinity was 35.9–36 g/kg. Temperature and salinity values at the two sites were similar (median difference 0.014°C and 0.001 g/kg respectively; two sample *t*-test, temperature p = 0.85, salinity p = 0.64).

3.3. Relationship Between pH and DO

pH and DO values correlated positively across all sites (Figure 7, Table 3), with distinct differences in regression characteristics between sites (Table 3). In Dominica, data from all five deployments were grouped together as there was no significant difference in pH:DO relationships between the deployments. There was a stronger correlation between pH and DO at Goff's Caye fore ($R^2 = 0.78$, p < 0.001) and back reef ($R^2 = 0.73$, p < 0.001) sites where the diel signal was also strongest (Figure 3, Table 3). At English Caye pH and DO correlation was weak ($R^2 = 0.30$, p < 0.001) but stronger when only the nighttime data were included ($R^2 = 0.58$, p < 0.001). Both the pH:DO correlation ($R^2 = 0.20$, p < 0.001) and diel signal were weakest at the Dominica site.

Classifying the data according to whether it was collected during daylight or in darkness highlighted distinctions in the pH:DO relationship, particularly at English Caye (Figure 7c, Table 3). At both Goff's Caye sites the pH:DO relationship was very similar between night and day and closely agreed with the theoretical P:R Model. Observations from English Caye agreed well with the theoretical P:R model during the night but not during daytime. At the Dominica site, there was no significant distinction between observations made during day or night.

3.4. Benthic Coverage

The four sites exhibited different benthic community compositions (Figure 8). Hard coral cover was highest at Goff's Caye fore reef and was significantly lower at the other sites. Goff's Caye fore reef was dominated by hard and soft corals and the back reef by sand, rubble and soft coral. English Caye's main benthos was macroalgae and soft coral, and Soufriere was dominated by rock and encrusting sponge, coral, or coralline crustose algae (CCA). Differentiating between encrusting organisms (sponges, coral or CCA) was not always obvious; therefore, they have been grouped together to avoid misclassification.

Figure 3. Diel cycles in pH_T (left hand side) and DO (right hand side). pH is reported on the total scale. All deployments for each site are included. The solid black line shows the median pH/DO and nighttime is shown with gray shading. Measurements throughout the day for the four sites (a + b, purple squares) Goff's Caye fore reef, Belize; (c + d, pink circles) Goff's Caye back reef, Belize; (e + f, turquoise triangles) English Caye, Belize; (g + h, orange diamonds) Soufriere, Dominica (five deployments). The pH and DO data from Dominica are on a smaller scale to better illustrate the diel signal. The bottom two panels compare the pH_T (i) and DO (j) median lines for each site: Goff's Caye fore reef is dashed purple; Goff's Caye back reef is solid pink; English Caye is dotted turquoise; and Dominica is dotted dashed orange.

4. Discussion

Measuring pH at high temporal resolution is recommended by the global ocean acidification community for monitoring both long-term changes in average seawater acidity and its temporal variability (range). Examining the short-term variability in pH (over hours, days, seasons) is most relevant in establishing ocean acidification impacts on marine ecosystems (Widdicombe et al., 2023). This is because it reveals the range in pH and the extreme organisms are exposed to. Such information can be generated effectively using the autonomous observing platforms described in this study, which in addition to pH record DO, salinity and temperature, enabling a more comprehensive characterization of biogeochemical processes. The pH values recorded during our study (7.849–8.090; Table 2) across all sites in Belize and Dominica were generally below the global seawater average at the time of sampling (8.07 between 60°N and 60°S; Jiang et al., 2019). Minimum pH was measured at Goff's Caye fore reef site (7.849), similar to that reported for other Caribbean reefs (e.g., ~7.9 on reefs in Bermuda, Panama and Puerto Rico; Cyronak et al., 2020; Gray et al., 2012; Pezner et al., 2021), but higher than some reported Pacific reef minimums (e.g., Palmyra atoll < 7.85; Price et al., 2012 and Great Barrier Reef < 7.7; Shaw et al., 2012). Minimum pH values from high-resolution studies on coral reefs are rarely reported and therefore, it remains difficult to make comparisons across space and time. While the pH minimums we recorded were not unusually low, they will continue to decline as the ocean CO₂ content increases, with lower pH minimums across the diel cycle due to the reduced buffering capacity of seawater (Cai et al., 2011; Page et al., 2016).

The diel ranges in pH recorded (0.070 Goff's Caye fore reef, 0.066 Goff's Caye back reef, 0.047 English Caye and 0.027 Dominica) are similar to those recorded at other Caribbean reefs—for example, 0.08 mean in Puerto Rico (Gray et al., 2012), and 0.03–0.07 at Hog Reef, Bermuda (Cyronak et al., 2020; Pezner et al., 2021; Takeshita et al., 2018). However, diel pH ranges recorded at Pacific reefs are around an order of magnitude larger than those recorded at our sites or elsewhere in the Caribbean—for example, 0.24 at Line Islands, Pacific (Price et al., 2012), 0.6 on Molokai Reef, Hawaii (Yates & Halley, 2006) and 0.7 at Heron Island, Australia (Santos et al., 2011). Higher diel pH variability has previously been associated with increased probability of coral recruit survival (Dufault et al., 2012) and higher coral calcification rates (Chan & Connolly, 2013; Chan & Eggins, 2017; Enochs et al., 2018), factors often used as indicators of reef health (Smith et al., 2016). Therefore, declining pH diel ranges, due to phase shifts and changes in ecosystem composition in general, may further exacerbate coral decline by reducing recruitment and calcification through a positive feedback loop. The lower diel pH ranges reported here and by previous studies may reflect the vulnerability of Caribbean reefs to environmental change and their higher susceptibility to phase shifts (Mumby et al., 2021).

The DO ranges recorded at our sites 30.7 μ mol/L at Goff's Caye fore reef (30.0 μ mol/kg), 32.0 μ mol/L at Goff's Caye back reef (31.3 μ mol/kg), 11.9 μ mol/L at English Caye (11.6 μ mol/kg), and 7.6 μ mol/L in Dominica (7.4 μ mol/kg) are generally lower than those reported in other studies (e.g., 22–54 μ mol/kg in Bermuda (Pezner et al., 2021; Takeshita et al., 2018) and in Puerto Rico 60 μ mol/L (fall) and 120 μ mol/L (winter) (Gray et al., 2012)). More recently, mean daily DO concentrations and ranges from 32 reefs around the globe have been reported as 173 ± 28 μ mol/kg and 81 ± 52 μ mol/kg respectively (Pezner et al., 2023). The greater DO amplitudes reported in the literature do not always correspond to greater pH ranges (0.067 (winter) and 0.093 (fall) in Puerto Rico (Gray et al., 2012) and 0.07 at Hog Reef (Pezner et al., 2021; Takeshita et al., 2018)) and could imply different biogeochemical and metabolic regimes, reflecting variations in benthic ecosystem compositions. Minimum DO values recorded at English Caye and Goff's Caye (both fore reef and back reef sites) were on occasion below 156 μ mol/L, a threshold below which reef ecosystems are considered "weakly hypoxic" and therefore under stress (Vaquer-Sunyer & Duarte, 2008). Climate change scenarios predict that warming oceans will further decrease minimum DO concentrations and lengthen the duration reefs are exposed to hypoxic conditions (Pezner et al., 2023).

The Goff's Caye observations illustrate how small-scale variability in pH within a single reef system can be determined using high-quality, high-resolution sensor measurements. Our measurements suggest that the back reef environment may be more biologically active, as evident by the higher median pH and DO concentrations (Table 2). This could be driven by closer proximity to terrestrial influence (e.g., nutrient inputs), as it has been suggested further south on the Mesoamerican Reef in Belize (Sapodilla Cayes) based on coral core isotopic data (Fowell et al., 2018). Fowell et al. (2018) data suggested that net ecosystem production and pH were higher in the part of the reef more exposed to terrestrial inputs (i.e., nutrients and sediments). The closer proximity of seagrass beds to the back reef site may also have been responsible for the higher pH and DO concentrations. Seagrass beds can significantly elevate seawater pH (>0.1) through the efficient consumption of seawater CO_2 (Ricart et al., 2021), release of TA from sediments (Burdige & Zimmerman, 2002; Palacios & Zimmerman, 2007) and

amplifying metabolic signals by increasing water residence time (Hendriks et al., 2015). Our data suggest that benthic community composition may have an effect on the amplitude of biogeochemical signals. Specifically, pH and DO diel ranges appear to correlate positively with hard coral cover with the fore reef site at Goff's Caye having the highest percentage of hard coral cover and the Dominica site the lowest. The diel DO and pH range recorded at English Caye were lower than those recorded at the Goff's Cayes sites (Figure 3), potentially reflecting the lower hard coral cover and higher macroalgae abundance (Figure 8). Despite the higher macroalgae abundance, median pH and DO diel ranges were lower than those at the Goff's Caye sites, in line with what has been observed during phase shifts from calcifying to fleshy species (e.g., Price et al., 2012). Price et al. (2012) found that declining median pH values measured in reefs of Palmyra Atoll corresponded with an increase in fleshy algae. In Dominica, the small amplitudes of pH and DO (Figures 3g and 3h) may be evidence of the relatively low biomass and live coral abundance (only 5% of each hard and soft coral) and the relatively high abundance of rock.

Although our data suggest a link between diel ranges (pH and DO) and benthic community composition and metabolism, the effect of topography and hydrology cannot be dismissed. The magnitude of biogeochemical signals can be a function of reef morphology (width, depth and wave/current characteristics (e.g., Cyronak et al., 2020; Falter et al., 2013; Pezner et al., 2023; Takeshita et al., 2018)). Therefore, it cannot be ruled out that some of the pH and DO diel range variability between our sites, and others reported in the literature, is due to factors additional to ecosystem metabolism. It is possible that the smaller diel cycles, observed at the Dominica site (2.5–4.5 m deeper) compared to those in Belize, may be driven in part by higher levels of mixing and dilution driven by the deeper and steeper topography. Significantly higher diel rages in pH and DO, however, have been recorded in reefs much deeper than our site in Dominica (depth (d) = 9 m, range = 0.027 pH and 7.4 μ mol/kg DO); in Bermuda, for example, (d = 18.5 m, range = 0.056 pH and 42 μ mol/kg DO; Takeshita et al., 2018) and in American Samoa (d = 15.2 m, range = 23 μ mol/kg DO range; Pezner et al., 2023). Consequently, using diel ranges as a proxy of benthic composition and metabolism may not be as meaningful if topography and hydrology are not taken into consideration.

The theoretical P:R model used here provides a means for interpreting the observed relationship between pH and DO measurements. In marine systems where photosynthesis and respiration dominate the control of DO, and dissolved CO_2 can be expressed by $C_6H_{12}O_6 + 6O_2 \leftrightarrow 6CO_2 + 6H_2O$. At a fixed TA, pH correlates negatively with dissolved CO_2 according to the speciation of carbonic acid in seawater (Zeebe & Wolf-Gladrow, 2003). Therefore, a positive correlation between DO and pH can be expected in productive marine habitats such as kelp forests (Frieder et al., 2012), seagrass beds (Hendriks et al., 2014), and coral reef systems (Gray et al., 2012; Pezner et al., 2021). Divergence from



Figure 5. Time series of English Caye of (a) pH_T, (b) DO, (c) temperature, (d) absolute salinity and (e) predicted tides over 6.5 weeks. Nighttime is shown by gray shading.



Global Biogeochemical Cycles

10.1029/2022GB007577



Figure 6. Time series of five deployments in Dominica, with night shown in gray shading; (a-c) are deployments in 2018 and (d-e) in 2019. pH_T is the solid green line and DO is the dotted blue line. Salinity and temperature data are in Supporting Information S1 (Figure S1).

the predicted P:R model could indicate that other physical, chemical, or biological processes may be at play (besides photosynthesis and respiration). In coral reef environments, for example, calcification or dissolution would result in a pH decrease or increase, respectively, without any effect on the DO concentration. The pH and DO observations at both Goff's Caye sites correlated well and were generally in good agreement with the theoretical P:R relationship throughout a 24 hr cycle. This suggests that significant biological activity (i.e., photosynthesis and respiration) was mainly responsible for the observed diel signals (Figure 7, Table 3). In contrast, the weaker pH:DO relationship and the relatively small observed pH and DO diel range (0.027 and 7.6 µmol/L, respectively, Figures 3 and 6) at the Dominica site likely reflect the relatively low ecosystem biomass (Figure 8) and therefore weak biological activity.



Figure 7. Relationship between pH and DO at all monitoring sites, with nighttime measurements represented by filled symbols and daytime by empty symbols. The solid black line represents the theoretical pH:DO relationship based on the PR model described in Section 2.5. The solid colored line is the linear fit of all data points at that site. (a) Goff's Caye Fore Reef, Belize, (b) Goff's Caye Back Reef, Belize, (c) English Caye, Belize and (d) Soufriere, Dominica.



						Day Pearson's
		Pearson's correlation	Night equation of best		Day equation of best fit	correlation
Site	Equation of best fit line	coefficient	fit line		line	coefficient
Goff's Caye Fore Reef, Belize	y = 0.00230x + 7.53	$0.8841 \ p < 0.0001$	y = 0.00293x + 7.42	$0.8894 \ p < 0.0001$	y = 0.00203x + 7.58	0.8789 p < 0.0001
Goff's Caye Back Reef, Belize	y = 0.00228x + 7.54	$0.8542 \ p < 0.0001$	y = 0.00245x + 7.51	$0.7857 \ p < 0.0001$	y = 0.00203x + 7.59	$0.8275 \ p < 0.0001$
English Caye, Belize	y = 0.00321x + 7.41	$0.5439 \ p < 0.0001$	y = 0.00195x + 7.62	$0.7643 \ p < 0.0001$	y = 0.00372x + 7.33	$0.4301 \ p < 0.0001$
Soufrière, Dominica	y = 0.00522x + 6.97	$0.4525 \ p < 0.0001$	y = 0.00475x + 7.06	$0.3512 \ p < 0.0001$	y = 0.00538x + 6.94	$0.4967 \ p < 0.0001$

Note. Best fit lines of all data are plotted on equivalent plots in Figure 7.

Decoupling between pH and DO at English Caye during daylight hours suggests that, unlike the other sites, photosynthesis was not always the primary control on the pH:DO relationship (Figure 7c). This is also evident in Figure 5, where pH and DO variability often appears to be decoupled. The disproportional increase in pH observed could potentially be caused by an increase in TA through water mass mixing or dissolution of carbonates. Water mass mixing can likely be ruled out, as there was no discernible correlation between temperature and salinity as one would expect from a two-endmember mixing scenario (Pearson's Correlation = -0.4, p < 0.0001). Similarly, lack of correlation between tidal cycles or tidal amplitude (predicted estimates for Belize City as illustrated on Figure 5; Tides4Fishing, 2019) and the relationship between pH and DO suggest that is unlikely that tide-induced TA release from porewaters (e.g., Faber et al., 2014) was significant or responsible for the disproportional daytime increase in pH. The exclusive occurrence of the pH:DO decoupling during daylight hours may point toward a light-dependent process.

The English Cave site had the largest relative distribution of macroalgae (Figure 8). Macroalgae are efficient photosynthesizers that can often outcompete corals for space and reduce coral recruitment (Birrell et al., 2008) leading to coral-to-algae phase shifts. Macroalgae have also been known to have allelopathic relationships with corals, either directly (Del Monaco et al., 2017; Rasher & Hay, 2010; Slattery & Lesser, 2014) or indirectly, via their associated microbial communities (Smith et al., 2006). Concentrations of allelopathic compounds and their affects are likely a function of macroalgae abundance. Allelopathic compounds released from macroalgae (such as lipid metabolites) have been shown to adversely affect corals through direct contact with coral surfaces, often leading to bleaching (Rasher & Hay, 2010; Rasher et al., 2011). Their release during photosynthesis (Paul & Ritson-Williams, 2008; Shalaby, 2011) can also indirectly affect corals by acting as a substrate to microbial communities and thereby influencing their growth and functioning. Macroalgae are efficient consumers of CO₂ maintaining high pH in the water column during the day (Smith et al., 2013). However, allelopathic effects on the microbial communities can enhance daytime respiration, which may lower the pH in microenvironments on coral or rubble surfaces and crevices promoting dissolution (Islam et al., 2016; Romanó de Orte et al., 2021). Although we have no direct evidence of such processes taking place at English Caye, allelopathy-mediated coral dissolution could potentially explain the observed daytime decoupling of pH:DO. Daytime dissolution is unusual as coral dissolution usually occurs at night when the absence of photosynthesis allows respiration to drive down pH and aragonite saturation state (e.g., Cyronak et al., 2018; Romanó de Orte et al., 2021; Webb et al., 2021). However, the dissolution of dead corals at seawater aragonite saturation states of >1 have been observed, proposed to be mediated by microbial respiration (Romanó de Orte et al., 2021), indicating that daytime dissolution is possible.

The pH:DO decoupling observed at English Caye could also be explained if oxygen ebullition was at play. Oxygen ebullition, the production of oxygen bubbles (Koschorreck et al., 2017), can occur as water in contact with algal surfaces becomes supersaturated with respect to O_2 (Howard et al., 2018). Macroalgal surfaces have been known to promote oxygen ebullition through heterogenous nucleation on algae surfaces (Silveira et al., 2019). Oxygen ebullition can lead to an underestimation of the seawater DO concentration by up to 20%, as most of the produced oxygen remains in gaseous form (Howard et al., 2018; Koschorreck et al., 2017). A 20% increase in the English Caye DO data (Figure 7c) would account for the observed pH:DO decoupling as illustrated in Supporting Information S1 (Figure S4).

To better understand what drives the pH:DO decoupling observed at English Caye, the seawater carbonate system would have to be fully constrained at high resolution. This would only be possible by measuring another carbonate system parameter such as TA or DIC. Doing so, dissolution and calcification could be estimated and changes in pH could be directly attributed to biotic or non-biotic processes. Although autonomous pH sensors are commercially available, sensors for measurements of TA or DIC are still in development (e.g., Schaap et al., 2021). Autonomous characterization of the carbonate system at high resolution would allow us to understand, quantify and compare key metabolic processes across different reefs such as net ecosystem dissolution and calcification and can enable long-term baseline monitoring of the coral reefs "metabolic pulse" (Cyronak et al., 2018).

High resolution reef monitoring over several diel and seasonal cycles is only possible using autonomous sensors. We have demonstrated that monitoring pH and DO (along with temperature and salinity) can offer insight into reef metabolism. A more holistic understanding of coral reef biogeochemistry and reef ecosystem functioning could be ascertained by constraining the carbonate system by measuring an addi-

Table 3

Statistical Relationships Between DO and pH at the Four Sites, All Data and Then Separated Into Night and Day Time Data





Figure 8. Benthic coverage of Goff's Caye fore reef, Goff's Caye back reef, and English Caye and Soufriere sites.

tional carbonate chemistry parameter such as TA or DIC, and including sensors for constraining primary production such as photosynthetically active radiation meters and fluorometers (for chlorophyll estimations), and constraining hydrodynamics such as acoustic Doppler current profilers. Practical considerations however, related to power availability, cost and site accessibility may be the determining factors on the number of parameters that can be monitored.

4.1. Conclusions

The loss and deterioration of coral reef ecosystems globally threaten global biodiversity, ecosystem function and the blue economy of coastal regions. In the Caribbean, over half of the population live within 1.5 km of the coastline (Schnitter et al., 2019), making reef decline an urgent and critical problem. Here we demonstrate that high resolution data of pH and DO offer insight into the functioning of coral reef ecosystems. Specifically, our data suggest a correlation between the strength and amplitude of the pH:DO relationship "the metabolic pulse" and benthic ecosystem composition. The strength of the metabolic pulse appeared to be a function of the balance between hard and soft corals, macroalgae and sponge abundance as well as proximity to other ecosystems such as seagrass beds. This was illustrated by comparisons of adjacent reef sites with different benthic community compositions (Goff's Caye and English Caye) which exhibited distinctly different DO and pH relationships and diel ranges. The weakest "metabolic pulse" as indicated by small pH and DO diel ranges and poor pH:DO correlation was recorded at the Dominica reef site where the ecosystem was heavily affected by Hurricane Maria. The high-resolution data also revealed unexpected biogeochemical functioning such as the decoupling of pH and DO during the day (at English Caye) which could be attributed to physical (oxygen ebullition) or geochemical (carbonate dissolution) processes which are still not well understood within the context of coral reef ecology and biogeochemistry.

Conflict of Interest

The authors declare no conflicts of interest relevant to this study.

Data Availability Statement

All sensor data are available from The University of Southampton Data Repository https://doi.org/10.5258/ SOTON/D2346. This data set includes pH_T , dissolved O_2 , temperature and salinity data from all four sites. At the same DOI, benthos data is included—the proportions of each benthos and substrate from the four sites, determined using Coral Point Count.



Acknowledgments

This work was supported by the United Kingdom Government through the Commonwealth Marine Economies Programme, which aims to enable safe and sustainable marine economies across Commonwealth Small Island Developing States. S.C. was supported by the NEXUSS Studentship Natural Environment Research Council (Grant NE/N012070/1) and NERC National Capability programme CLASS (Climate Linked Atlantic Sector Science), Grant NE/R015953/1. C.E. was supported by Natural Environment Research Council (NERC) Independent Research Fellowship NE/M018806/1. We would like to thank both our Belizean (Coastal Zone Management and Authority Belize and The Belize Port Authority) and Dominica Partners (Dominica Fisheries Division and Natura Island Dive) for their warm welcome and enthusiasm for the project. We would additionally like to thank Urska Martinicic (NOC) for deployment assistance, David Paxton (NOC) for design and manufacture of the platform, Anna Cryer for statistical advice, Duncan Stevens (UoS) for proof reading and Veerle Huvenne (NOC) for input on an early draft. Finally, we would like to thank our anonymous reviewers and the associated editor for constructive reviews, which greatly improved the manuscript.

References

- Alves, C., Valdivia, A., Aronson, R. B., Bood, N., Castillo, K. D., Cox, C., et al. (2022). Twenty years of change in benthic communities across the Belizean Barrier Reef. *Plos One*, 17(1), e0249155. https://doi.org/10.1371/journal.pone.0249155
- Anderson, L. A. (1995). On the hydrogen and oxygen content of marine phytoplankton. Deep Sea Research Part I: Oceanographic Research Papers, 42(9), 1675–1680. https://doi.org/10.1016/0967-0637(95)00072-e
- Andersson, A. J., & Gledhill, D. (2013). Ocean acidification and coral reefs: Effects on breakdown, dissolution, and net ecosystem calcification. *Annual Review of Marine Science*, 5(1), 321–348. https://doi.org/10.1146/annurev-marine-121211-172241
- Anthony, K. R. N., Kleypas, J. A., & Gattuso, J. P. (2011). Coral reefs modify their seawater carbon chemistry—Implications for impacts of ocean acidification. *Global Change Biology*, 17(12), 3655–3666. https://doi.org/10.1111/j.1365-2486.2011.02510.x

Antonius, A., & Ballesteros, E. (1998). Epizoism: A new threat to coral health in Caribbean reefs. Revista de Biologia Tropical, 46, 145-156.

- Bennett, H. M., Altenrath, C., Woods, L., Davy, S. K., Webster, N. S., & Bell, J. J. (2017). Interactive effects of temperature and pCO₂ on sponges: From the cradle to the grave. *Global Change Biology*, 23(5), 2031–2046. https://doi.org/10.1111/gcb.13474
- Birrell, C. L., McCook, L. J., Willis, B. L., & Diaz-Pulido, G. A. (2008). Effects of benthic algae on the replenishment of corals and the implications for the resilience of coral reefs. *Oceanography and Marine Biology an Annual Review*, 46, i–iv.
- Bove, C. B., Ries, J. B., Davies, S. W., Westfield, I. T., Umbanhowar, J., & Castillo, K. D. (2019). Common Caribbean corals exhibit highly variable responses to future acidification and warming. *Proceedings of the Royal Society B*, 286(1900), 20182840. https://doi.org/10.1098/ rspb.2018.2840
- Bove, C. B., Whitehead, R. F., & Szmant, A. M. (2020). Responses of coral gastrovascular cavity pH during light and dark incubations to reduced seawater pH suggest species-specific responses to the effects of ocean acidification on calcification. *Coral Reefs*, 39(6), 1675–1691. https:// doi.org/10.1007/s00338-020-01995-7
- Bozec, Y. M., O'Farrell, S., Bruggemann, J. H., Luckhurst, B. E., & Mumby, P. J. (2016). Tradeoffs between fisheries harvest and the resilience of coral reefs. Proceedings of the National Academy of Sciences of the United States of America, 113(16), 4536–4541. https://doi.org/10.1073/ pnas.1601529113
- Brooker, R. M., Hay, M. E., & Dixson, D. L. (2016). Chemically cued suppression of coral reef resilience: Where is the tipping point? Coral Reefs, 35(4), 1263–1270. https://doi.org/10.1007/s00338-016-1474-4
- Burdige, D. J., & Zimmerman, R. C. (2002). Impact of sea grass density on carbonate dissolution in Bahamian sediments. Limnology & Oceanography, 47(6), 1751–1763. https://doi.org/10.4319/lo.2002.47.6.1751
- Cai, W. J., Hu, X., Huang, W. J., Jiang, L. Q., Wang, Y., Peng, T. H., & Zhang, X. (2010). Alkalinity distribution in the western North Atlantic Ocean margins. *Journal of Geophysical Research*, 115(C8), 1–15. https://doi.org/10.1029/2009jc005482
- Cai, W. J., Hu, X., Huang, W. J., Murrell, M. C., Lehrter, J. C., Lohrenz, S. E., et al. (2011). Acidification of subsurface coastal waters enhanced by eutrophication. *Nature Geoscience*, 4(11), 766–770. https://doi.org/10.1038/ngeo1297
- Chan, N. C. S., & Connolly, S. R. (2013). Sensitivity of coral calcification to ocean acidification: A meta-analysis. *Global Change Biology*, 19(1), 282–290. https://doi.org/10.1111/gcb.12011
- Chan, W. Y., & Eggins, S. M. (2017). Calcification responses to diurnal variation in seawater carbonate chemistry by the coral Acropora formosa. Coral Reefs, 36(3), 763–772. https://doi.org/10.1007/s00338-017-1567-8
- Clayton, T. D., & Byrne, R. H. (1993). Spectrophotometric seawater pH measurements: Total hydrogen ion concentration scale calibration of m-cresol purple and at-sea results. *Deep Sea Research Part I: Oceanographic Research Papers*, 40(10), 2115–2129. https://doi. org/10.1016/0967-0637(93)90048-8
- Cox, C., Valdivia, A., McField, M., Castillo, K., & Bruno, J. F. (2017). Establishment of marine protected areas alone does not restore coral reef communities in Belize. *Marine Ecology Progress Series*, 563, 65–79. https://doi.org/10.3354/meps11984
- Cryer, S., Carvalho, F., Wood, T., Strong, J. A., Brown, P., Loucaides, S., et al. (2020). Evaluating the sensor-equipped autonomous surface vehicle c-worker 4 as a tool for identifying coastal ocean acidification and changes in carbonate chemistry. *Journal of Marine Science and Engineering*, 8(11), 1–18. https://doi.org/10.3390/jmse8110939
- Cyronak, T., Andersson, A. J., Langdon, C., Albright, R., Bates, N. R., Caldeira, K., et al. (2018). Taking the metabolic pulse of the world's coral reefs. *PLoS One*, 13, 1–17. https://doi.org/10.1371/journal.pone.0190872
- Cyronak, T., Takeshita, Y., Courtney, T. A., DeCarlo, E. H., Eyre, B. D., Kline, D. I., et al. (2020). Diel temperature and pH variability scale with depth across diverse coral reef habitats. *Limnology and Oceanography Letters*, 5(2), 193–203. https://doi.org/10.1002/lol2.10129
- Del Monaco, C., Hay, M. E., Gartrell, P., Mumby, P. J., & Diaz-Pulido, G. (2017). Effects of ocean acidification on the potency of macroalgal allelopathy to a common coral. *Scientific Reports*, 7, 1–10. https://doi.org/10.1038/srep41053
- Dickson, A. G. (1990). Standard potential of the reaction: AgCl(s) + 12H2(g) = Ag(s) + HCl(aq) and the standard acid constants of the ions HSO4- in synthetic seawater from 273.15 to 318.15 K. *Journal of Chemical Thermodynamics*, 22(2), 113–227. https://doi.org/10.1016/0021-9614(90)90074-z
- Dickson, A. G. (2010). Standards for ocean measurements. Oceanography, 23(3), 34–47. https://doi.org/10.5670/oceanog.2010.22
- Dickson, A. G., Afghan, J. D., & Anderson, G. C. (2003). Reference materials for oceanic CO₂ analysis: A method for the certification of total alkalinity. *Marine Chemistry*, 80(2–3), 185–197. https://doi.org/10.1016/s0304-4203(02)00133-0
- Dickson, A. G., & Millero, F. J. (1987). A comparison of the equilibrium constants for the dissociation of carbonic acid in seawater media. Deep Sea Research Part A. Oceanographic Research Papers, 34(10), 1733–1743. https://doi.org/10.1016/0198-0149(87)90021-5
- D'Olivo, J. P., & McCulloch, M. T. (2017). Response of coral calcification and calcifying fluid composition to thermally induced bleaching stress. *Scientific Reports*, 7, 1–15. https://doi.org/10.1038/s41598-017-02306-x
- Dufault, A. M., Cumbo, V. R., Fan, T. Y., & Edmunds, P. J. (2012). Effects of diurnally oscillating pCO₂ on the calcification and survival of coral recruits. *Proceedings of the Royal Society B: Biological Sciences*, 279(1740), 2951–2958. https://doi.org/10.1098/rspb.2011.2545
- Eddy, T. D., Lam, V. W. Y., Reygondeau, G., Cisneros-Montemayor, A. M., Greer, K., Palomares, M. L. D., et al. (2021). Global decline in capacity of coral reefs to provide ecosystem services. *One Earth*, 4(9), 1278–1285. https://doi.org/10.1016/j.oneear.2021.08.016
- Enochs, I. C., Manzello, D. P., Jones, P. J., Aguilar, C., Cohen, K., Valentino, L., et al. (2018). The influence of diel carbonate chemistry fluctuations on the calcification rate of Acropora cervicornis under present day and future acidification conditions. *Journal of Experimental Marine Biology and Ecology*, 506, 135–143. https://doi.org/10.1016/j.jembe.2018.06.007
- Erez, J., Reynaud, S., Silverman, J., Schneider, K., & Allemand, D. (2011). Coral calcification under ocean acidification and global change. In Z. Dubinsky, & N. Stambler (Eds.), Coral reefs: An ecosystem in transition (pp. 151–176). Springer Netherlands.
- Faber, P. A., Evrard, V., Woodland, R. J., Cartwright, I. C., & Cook, P. L. M. (2014). Pore-water exchange driven by tidal pumping causes alkalinity export in two intertidal inlets. *Limnology & Oceanography*, 59(5), 1749–1763. https://doi.org/10.4319/lo.2014.59.5.1749

Falter, J. L., Lowe, R. J., Zhang, Z., & McCulloch, M. (2013). Physical and biological controls on the carbonate chemistry of coral reef waters: Effects of metabolism, wave forcing, sea level, and geomorphology. *PLoS One*, *8*(1), e53303. https://doi.org/10.1371/journal.pone.0053303 Fowell, S. E., Foster, G. L., Ries, J. B., Castillo, K. D., de la Vega, E., Tyrrell, T., et al. (2018). Historical trends in pH and carbonate biogeochemis-

try on the Belize Mesoamerican Barrier Reef System *Geophysical Research Letters*, 45(7), 3228–3237. https://doi.org/10.1002/2017gl076496
Frieder, C. A., Nam, S. H., Martz, T. R., & Levin, L. A. (2012). High temporal and spatial variability of dissolved oxygen and pH in a nearshore California kelp forest. *Biogeosciences*, 9(10), 3917–3930. https://doi.org/10.5194/be-9-3917-2012

Gardner, T. A., Côté, I. M., Gill, J. A., Grant, A., & Watkinson, A. R. (2003). Long-term region-wide declines in Caribbean corals. *Science*, 301(5635), 958–960. https://doi.org/10.1126/science.1086050

Gattuso, J.-P., Allemand, D., & Frankignoulle, M. (1999). Photosynthesis and calcification at cellular, organismal and community levels in coral reefs: A review on interactions and control by carbonate chemistry. *American Zoologist*, 39(1), 160–183. https://doi.org/10.1093/icb/39.1.160

Google Earth Pro V 7.3.6.9345. (2011). Goff's Caye, Belize. 17.380716, -88.074173, Eye alt 11.6 km. SIO, NOAA, U.S. Navy, NGA, GEBCO. Maxar Technologies 2020. [19th March 2023].

- Google Earth Pro V 7.3.6.9345. (2015a). Caribbean Sea. 18.954288, -73.75448, Eye alt 3249.4 km. SIO, NOAA, U.S. Navy, NGA, GEBCO, LDEO-Columbia, NSF. Landset/Corpicanus, Maxar Technologies 2023. [19th March 2023].
- Google Earth Pro V 7.3.6.9345. (2015b). Dominica. 15.372298, -61.267555 Eye alt 100.7 km. SIO, NOAA, U.S. Navy, NGA, GEBCO, CNES/ Airbus 2023, Maxar Technologies 2020. [3rd April 2020].
- Google Earth Pro V 7.3.6.9345. (2015c). Soufriere, Dominica. 15.230794, -61.350874, Eye alt 11 km. SIO, NOAA, U.S. Navy, NGA, GEBCO, CNES/Airbus 2020. [26th November 2020].

Google Earth Pro V 7.3.6.9345. (2019). Belize City, Belize. 17.434539, -88.238827, Eye alt 50.3 km. SIO, NOAA, U.S. Navy, NGA, GEBCO. CNES/Airbus 2020, Maxar Technologies 2020. [3rd April 2020].

- Gray, S. E. C., Degrandpre, M. D., Langdon, C., & Corredor, J. E. (2012). Short-term and seasonal pH. *Global Biogeochemical Cycles*, 26, 1–13. https://doi.org/10.1029/2011GB004114
- Hendriks, I. E., Duarte, C. M., Olsen, Y. S., Steckbauer, A., Ramajo, L., Moore, T. S., et al. (2015). Biological mechanisms supporting adaptation to ocean acidification in coastal ecosystems. *Estuarine, Coastal and Shelf Science*, 152, A1–A8. https://doi.org/10.1016/j.ecss.2014.07.019
- Hendriks, I. E., Olsen, Y. S., Ramajo, L., Basso, L., Steckbauer, A., Moore, T. S., et al. (2014). Photosynthetic activity buffers ocean acidification in seagrass meadows. *Biogeosciences*, 11(2), 333–346. https://doi.org/10.5194/bg-11-333-2014
- Hoegh-Guildberg, O. E., Mumby, P. J. J., Hooten, A. J. J., Steneck, R. S. S., Greenfield, P., Gomez, E., et al. (2007). Coral reefs under rapid climate change and ocean acidification. *Science*, 318(5857), 1737–1742. https://doi.org/10.1126/science.1152509
- Howard, E. M., Forbrich, I., Giblin, A. E., Lott, D. E., Cahill, K. L., & Stanley, R. H. R. (2018). Using noble gases to compare parameterizations of air-water gas exchange and to constrain oxygen losses by ebullition in a shallow aquatic environment. *Journal of Geophysical Research: Biogeosciences*, 123(9), 2711–2726. https://doi.org/10.1029/2018jg004441
- Hughes, T. P. (1994). Catastrophes, phase shifts, and large-scale degradation of a Caribbean coral reef. Science, 265(5178), 1547–1550. https://doi.org/10.1126/science.265.5178.1547
- Hughes, T. P., Graham, N. A. J., Jackson, J. B. C., Mumby, P. J., & Steneck, R. S. (2010). Rising to the challenge of sustaining coral reef resilience. *Trends in Ecology & Evolution*, 25(11), 633–642. https://doi.org/10.1016/j.tree.2010.07.011
- Islam, M. N., Casareto, B. E., & Suzuki, Y. (2016). Eutrophication accelerates carbonate dissolution under high pCO₂ condition in coral reef ecosystem. *Journal of Scientific Research*, 8(3), 427–438. https://doi.org/10.3329/jsr.v8i3.27937
- Jiang, L. Q., Carter, B. R., Feely, R. A., Lauvset, S. K., & Olsen, A. (2019). Surface ocean pH and buffer capacity: Past, present and future. Scientific Reports, 9, 1–11. https://doi.org/10.1038/s41598-019-55039-4
- Jokiel, P. L., Jury, C. P., & Rodgers, K. S. (2014). Coral-algae metabolism and diurnal changes in the CO₂ -carbonate system of bulk sea water. *PeerJ*, 2, e378. https://doi.org/10.7717/peerj.378
- Kleypas, J. A., Buddemeier, R. W., Archer, D., Gattuso, J., Langdon, C., & Opdyke, B. N. (1999). Geochemical consequences of increased atmospheric carbon dioxide on coral reefs. *Science*, 284(5411), 118–120. https://doi.org/10.1126/science.284.5411.118
- Kline, D. I., Teneva, L., Hauri, C., Schneider, K., Miard, T., Chai, A., et al. (2015). Six month in situ high-resolution carbonate chemistry and temperature study on a coral reef flat reveals asynchronous pH and temperature anomalies. *PloS One*, 10(6), 1–26. https://doi.org/10.1371/ journal.pone.0127648
- Kohler, K. E., & Gill, S. M. (2006). Coral point count with excel extensions (CPCe): A visual basic program for the determination of coral and substrate coverage using random point count methodology. *Computers & Geosciences*, 32(9), 1259–1269. https://doi.org/10.1016/j. cageo.2005.11.009
- Koschorreck, M., Hentschel, I., & Boehrer, B. (2017). Oxygen ebullition from lakes. Geophysical Research Letters, 44(18), 9372–9378. https:// doi.org/10.1002/2017g1074591
- Langdon, C., Broecker, W. S., Hammond, D. E., Glenn, E., Fitzsimmons, K., Nelson, S. G., et al. (2003). Effect of elevated CO₂ on the community metabolism of an experimental coral reef. *Global Biogeochemical Cycles*, *17*(1). https://doi.org/10.1029/2002gb001941
- Lee, K., Kim, T.-W., Byrne, R. H., Millero, F. J., Feely, R. A., & Liu, Y.-M. (2010). The universal ratio of boron to chlorinity for the North Pacific and North Atlantic Oceans. *Geochimica et Cosmochimica Acta*, 74(6), 1801–1811. https://doi.org/10.1016/j.gca.2009.12.027

Lewis, E. R., & Wallace, D. W. R. (1998). Program developed for CO₂ system calculations web. Liu, X., Patsavas, M. C., & Byrne, R. H. (2011). Purification and characterization of meta-cresol purple for spectrophotometric seawater pH

measurements. Environmental Science & Technology, 45(11), 4862–4868. https://doi.org/10.1021/es200665d
López-Victoria, M., & Zea, S. (2005). Current trends of space occupation by encrusting excavating sponges on Colombian coral reefs. Marine Ecology, 26(1), 33–41. https://doi.org/10.1111/j.1439-0485.2005.00036.x

Madisetti, A. (2018). Marine damage report & dive sector needs assessment Commonwealth of Dominica.

Matlab. (2018). MATLAB version: 9.5.0 (2018b). The MathWorks Inc. Retrieved from https://www.mathworks.com

Matlab. (2022). MATLAB version: 9.13.0 (R2022b). The MathWorks Inc. Retrieved from https://www.mathworks.com

McMahon, A., Santos, I. R., Schulz, K. G., Cyronak, T., & Maher, D. T. (2018). Determining coral reef calcification and primary production using automated alkalinity, pH and pCO₂ measurements at high temporal resolution. *Estuarine, Coastal and Shelf Science*, 209, 80–88. https:// doi.org/10.1016/j.ecss.2018.04.041

- Mehrbach, C., Culberson, C. H., Hawley, J. E., & Pytkowicx, R. M. (1973). Measurement of the apparent dissociation constants of carbonic acid in seawater at atmospheric pressure. *Limnology & Oceanography*, *18*(6), 897–907. https://doi.org/10.4319/lo.1973.18.6.0897
- Mintrop, L. (2004). VINDTA, versatile instrument for the determination of titration Alkalinity. Manual for versions 3S and 3C. Version 2.0. In *Marine Analytics and data (MARIANDA)* (p. 45).
- Moav-Barzel, O., Erez, J., Lazar, B., & Silverman, J. (2023). Higher nighttime rates of CaCO₃ dissolution in the nature reserve reef, Eilat, Israel in 2015–2016 compared to 2000–2002. Journal of Geophysical Research: Biogeosciences, 128, 1–22. https://doi.org/10.1029/2021jg006763

- Moya, A., Tambutté, S., Tambutté, E., Zoccola, D., Caminiti, N., & Allemand, D. (2006). Study of calcification during a daily cycle of the coral *Stylophora pistillata*: Implications for "light-enhanced calcification. *Journal of Experimental Biology*, 209(17), 3413–3419. https://doi. org/10.1242/jeb.02382
- Mumby, P. J., Hastings, A., & Edwards, H. J. (2007). Thresholds and the resilience of Caribbean coral reefs. Nature, 450(7166), 98–101. https:// doi.org/10.1038/nature06252
- Mumby, P. J., Steneck, R. S., Roff, G., & Paul, V. J. (2021). Marine reserves, fisheries ban, and 20 years of positive change in a coral reef ecosystem. (pp. 1–11).
- Norström, A. V., Nyström, M., Lokrantz, J., & Folke, C. (2009). Alternative states on coral reefs: Beyond coral-macroalgal phase shifts. *Marine Ecology Progress Series*, 376, 293–306. https://doi.org/10.3354/meps07815
- Nyström, M., Graham, N. A. J., Lokrantz, J., & Norström, A. V. (2008). Capturing the cornerstones of coral reef resilience: Linking theory to practice. Coral Reefs, 27(4), 795–809. https://doi.org/10.1007/s00338-008-0426-z
- Page, H. N., Andersson, A. J., Jokiel, P. L., Rodgers, K. S., Lebrato, M., Yeakel, K., et al. (2016). Differential modification of seawater carbonate chemistry by major coral reef benthic communities. *Coral Reefs*, 35(4), 1311–1325. https://doi.org/10.1007/s00338-016-1490-4
- Palacios, S., & Zimmerman, R. C. (2007). Response of eelgrass Zostera marina to CO₂ enrichment: Possible impacts of climate change and potential for remediation of coastal habitats. *Marine Ecology Progress Series*, 344, 1–13. https://doi.org/10.3354/meps07084
- Paul, V. J., & Ritson-Williams, R. (2008). Marine chemical ecology. *Natural Product Reports*, 25(4), 662–695. https://doi.org/10.1039/b702742gPendleton, L., Comte, A., Langdon, C., Ekstrom, J. A., Cooley, S. R., Suatoni, L., et al. (2016). Coral reefs and people in a high-CO₂ world: Where can science make a difference to people? *PloS One*, 11, 1–21. https://doi.org/10.1371/journal.pone.0164699
- Perez, F. F., & Fraga, F. (1987). Association constant of fluoride and hydrogen ions in seawater. *Marine Chemistry*, 21(2), 161–168. https://doi. org/10.1016/0304-4203(87)90036-3
- Perry, C. T., Alvarez-Filip, L., Graham, N. A. J., Mumby, P. J., Wilson, S. K., Kench, P. S., et al. (2018). Loss of coral reef growth capacity to track future increases in sea level. *Nature*, 558(7710), 396–400. https://doi.org/10.1038/s41586-018-0194-z
- Pezner, A. K., Courtney, T. A., Barkley, H. C., Chou, W.-C., Chu, H.-C., Clements, S. M., et al. (2023). Increasing hypoxia on global coral reefs under ocean warming. *Nature Climate Change*, 13(4), 403–409. https://doi.org/10.1038/s41558-023-01619-2
- Pezner, A. K., Courtney, T. A., Page, H. N., Giddings, S. N., Beatty, C. M., DeGrandpre, M. D., & Andersson, A. J. (2021). Lateral, vertical, and temporal variability of seawater carbonate chemistry at Hog Reef, Bermuda. *Frontiers in Marine Science*, 8, 1–18. https://doi.org/10.3389/ fmars.2021.562267
- Price, N. N., Martz, T. R., Brainard, R. E., & Smith, J. E. (2012). Diel variability in seawater pH relates to calcification and benthic community structure on coral reefs. *PloS One*, 7(8), 1–9. https://doi.org/10.1371/journal.pone.0043843
- Rasher, D. B., & Hay, M. E. (2010). Seaweed allelopathy degrades the resilience and function of coral reefs. *Communicative & Integrative Biology*, 3(6), 564–566. https://doi.org/10.4161/cib.3.6.12978
- Rasher, D. B., Stout, E. P., Engel, S., Kubanek, J., & Hay, M. E. (2011). Macroalgal terpenes function as allelopathic agents against reef corals. Proceedings of the National Academy of Sciences of the United States of America, 108(43), 17726–17731. https://doi.org/10.1073/ pnas.1108628108
- Ricart, A. M., Ward, M., Hill, T. M., Sanford, E., Kroeker, K. J., Takeshita, Y., et al. (2021). Coast-wide evidence of low pH amelioration by seagrass ecosystems. *Global Change Biology*, 1–12(11), 2580–2591. https://doi.org/10.1111/gcb.15594
- Romanó de Orte, M., Koweek, D. A., Cyronak, T., Takeshita, Y., Griffin, A., Wolfe, K., et al. (2021). Unexpected role of communities colonizing dead coral substrate in the calcification of coral reefs. *Limnology & Oceanography*, 1–11(5), 1793–1803. https://doi.org/10.1002/lno.11722 Roth, M. S. (2014). The engine of the reef: Photobiology of the coral-algal symbiosis. *Frontiers in Microbiology*, 5, 1–22.
- Santos, I. R., Glud, R. N., Maher, D., Erler, D., & Eyre, B. D. (2011). Diel coral reef acidification driven by porewater advection in permeable carbonate sands, Heron Island, Great Barrier Reef. Geophysical Research Letters, 38(3), 1–5. https://doi.org/10.1029/2010g1046053
- Schaap, A., Koopmans, D., Holtappels, M., Dewar, M., Arundell, M., Papadimitriou, S., et al. (2021). Quantification of a subsea CO₂ release with lab-on-chip sensors measuring benthic gradients. *International Journal of Greenhouse Gas Control*, 110, 103427. https://doi.org/10.1016/j. ijgcc.2021.103427
- Schnitter, R., Verret, M., Berry, P., Fook, T. C. T., Hales, S., Lal, A., & Edwards, S. (2019). An assessment of climate change and health vulnerability and adaptation in Dominica. *International Journal of Environmental Research and Public Health*, 16(1), 70. https://doi.org/10.3390/ ijerph16010070
- Shalaby, E. A. (2011). Algae as promising organisms for environment and health. *Plant Signaling & Behavior*, 6(9), 1338–1350. https://doi.org/10.4161/psb.6.9.16779
- Shamberger, K. E. F., Lentz, S. J., & Cohen, A. L. (2018). Low and variable ecosystem calcification in a coral reef lagoon under natural acidification. *Limnology & Oceanography*, 63(2), 714–730. https://doi.org/10.1002/lno.10662
- Sharp, J. D., Pierrot, D., Humphreys, M. P., Epitalon, J.-M., Orr, J. C., Lewis, E. R., & Wallace, D. W. R. (2020). CO2SYSv3 for MATLAB.
- Shaw, E. C., McNeil, B. I., & Tilbrook, B. (2012). Impacts of ocean acidification in naturally variable coral reef flat ecosystems. Journal of Geophysical Research, 117(C3), 1–14. https://doi.org/10.1029/2011jc007655
- Silbiger, N. J., Guadayol, O., Thomas, F. I. M. M., & Donahue, M. J. (2014). Reefs shift from net accretion to net erosion along a natural environmental gradient. *Marine Ecology Progress Series*, 515, 33–44. https://doi.org/10.3354/meps10999
- Silbiger, N. J., Nelson, C. E., Remple, K., Sevilla, J. K., Quinlan, Z. A., Putnam, H. M., et al. (2018). Nutrient pollution disrupts key ecosystem functions on coral reefs. Proceedings of the Royal Society B, 285(1880), 2–10. https://doi.org/10.1098/rspb.2017.2718
- Silveira, C. B., Luque, A., Roach, T. N., Villela, H., Barno, A., Green, K., et al. (2019). Biophysical and physiological processes causing oxygen loss from coral reefs. *Elife*, 8, 1–24. https://doi.org/10.7554/elife.49114
- Slattery, M., & Lesser, M. P. (2014). Allelopathy in the tropical alga Lobophora variegata (Phaeophyceae): Mechanistic basis for a phase shift on mesophotic coral reefs? Journal of Phycology, 50(3), 493–505. https://doi.org/10.1111/jpy.12160
- Smith, J. E., Brainard, R., Carter, A., Grillo, S., Edwards, C., Harris, J., et al. (2016). Re-evaluating the health of coral reef communities: Baselines and evidence for human impacts across the central pacific. *Proceedings of the Royal Society B: Biological Sciences*, 283(1822), 20151985. https://doi.org/10.1098/rspb.2015.1985
- Smith, J. E., Price, N. N., Nelson, C. E., & Haas, A. F. (2013). Coupled changes in oxygen concentration and pH caused by metabolism of benthic coral reef organisms. *Marine Biology*, 160(9), 2437–2447. https://doi.org/10.1007/s00227-013-2239-z
- Smith, J. E., Shaw, M., Edwards, R. A., Obura, D., Pantos, O., Sala, E., et al. (2006). Indirect effects of algae on coral: Algae-mediated, microbe-induced coral mortality. *Ecology Letters*, 9(7), 835–845. https://doi.org/10.1111/j.1461-0248.2006.00937.x
- Steiner, S. C. (2015). Coral Reefs of Dominica (Lesser Antilles). Annalen des Naturhistorischen Museums in Wien. Serie B für Botanik und Zoologie, 47–119.

- Steiner, Z., Turchyn, A. V., Harpaz, E., & Silverman, J. (2018). Water chemistry reveals a significant decline in coral calcification rates in the southern Red Sea. *Nature Communications*, 9(1), 3615. https://doi.org/10.1038/s41467-018-06030-6
- Stumm, W., & Morgan, J. J. (1996). Aquatic chemistry (3rd ed.). John Wiley & Sons.
- Suzuki, A., Nakamori, T., & Kayanne, H. (1995). The mechanism of production enhancement in coral reef carbonate systems: Model and empirical results. *Sedimentary Geology*, 99(3–4), 259–280. https://doi.org/10.1016/0037-0738(95)00048-d
- Takeshita, Y., Cyronak, T., Martz, T. R., Kindeberg, T., & Andersson, A. J. (2018). Coral reef carbonate chemistry variability at different functional scales. Frontiers in Marine Science, 5. https://doi.org/10.3389/fmars.2018.00175
- Tides4Fishing. (2019). Tide times and charts for Belize and weather forecast for fishing in Belize. [WWW Document]. Retrieved from https://tides4fishing.com/bz/belice/belize
- Vaquer-Sunyer, R., & Duarte, C. M. (2008). Thresholds of hypoxia for marine biodiversity. Proceedings of the National Academy of Sciences of the United States of America, 105(40), 15452–15457. https://doi.org/10.1073/pnas.0803833105
- Wanninkhof, R., Pierrot, D., Sullivan, K., Barbero, L., & Triñanes, J. (2020). A 17-year dataset of surface water fugacity of CO₂ along with calculated pH, aragonite saturation state and air-sea CO₂ fluxes in the northern Caribbean Sea. *Earth System Science Data*, 12(3), 1489–1509. https://doi.org/10.5194/essd-12-1489-2020
- Webb, A. E., De Bakker, D. M., Soetaert, K., Da Costa, T., Van Heuven, S. M. A. C., Van Duyl, F. C., et al. (2021). Quantifying functional consequences of habitat degradation on a Caribbean coral reef. *Biogeosciences*, 18(24), 6501–6516. https://doi.org/10.5194/bg-18-6501-2021
- Widdicombe, S., Isensee, K., Artioli, Y., Gaitán-Espitia, J. D., Hauri, C., Newton, J. A., et al. (2023). Unifying biological field observations to detect and compare ocean acidification impacts across marine species and ecosystems: What to monitor and why. Ocean Science, 19(1), 101–119. https://doi.org/10.5194/os-19-101-2023
- Wolf-Gladrow, D. A., Zeebe, R. E., Klaas, C., Körtzinger, A., & Dickson, A. G. (2007). Total alkalinity: The explicit conservative expression and its application to biogeochemical processes. *Marine Chemistry*, 106(1–2), 287–300. https://doi.org/10.1016/j.marchem.2007.01.006
- Yates, K. K., & Halley, R. (2006). CO₂- 3 concentration and pCO₂ thresholds for calcification and dissolution on the Molokai reef flat, Hawaii. *Biogeosciences*, *3*, 357–369. https://doi.org/10.5194/bg-3-357-2006
- Yin, T., Papadimitriou, S., Rérolle, V. M. C., Arundell, M., Cardwell, C. L., Walk, J., et al. (2021). A novel lab-on-chip spectrophotometric pH sensor for autonomous in situ seawater measurements to 6000 m depth on stationary and moving observing platforms. *Environmental Science* & Technology, 55(21), 14968–14978. https://doi.org/10.1021/acs.est.1c03517
- Zeebe, R. E., & Wolf-Gladrow, D. (2003). CO2 in seawater: Equilibrium, kinetics, isotopes (2nd ed.). Elsevier Science B.V.